

Risk factors and outcomes associated with recurrent autoimmune hepatitis following liver transplantation

Montano-Loza, A.J.; Ronca, V.; Ebadi, M.; Hansen, B.E.; Hirschfield, G.; Elwir, S.; ... ; Int Autoimmune Hepatitis Grp IAIHG

Citation

Montano-Loza, A. J., Ronca, V., Ebadi, M., Hansen, B. E., Hirschfield, G., Elwir, S., ... Oo, Y. H. (2022). Risk factors and outcomes associated with recurrent autoimmune hepatitis following liver transplantation. *Journal Of Hepatology*, *77*(1), 84-97. doi:10.1016/j.jhep.2022.01.022

Version:Publisher's VersionLicense:Creative Commons CC BY 4.0 licenseDownloaded from:https://hdl.handle.net/1887/3486070

Note: To cite this publication please use the final published version (if applicable).

Risk factors and outcomes associated with recurrent autoimmune hepatitis following liver transplantation

Graphical abstract



Highlights

- Recurrent AIH frequently occurs following LT.
- Recurrent AIH is associated with younger age at LT, use of mycophenolate mofetil post-LT, sex mismatch and higher IgG pre-LT.
- Recurrent AIH impacts on graft and overall survival after LT.
- Recurrent AIH following LT is clinically meaningful and requires improved management strategies.

Authors

Aldo J. Montano-Loza, Vincenzo Ronca, Maryam Ebadi, ..., Andrew L. Mason, Michael Heneghan , Ye Htun Oo

Correspondence

montanol@ualberta.ca (A.J. Montano-Loza), y.h.oo@bham. ac.uk (Y.H. Oo).

Lay summary

Recurrent autoimmune hepatitis following liver transplant is frequent and is associated with some recipient features and the type of immunosuppressive medications use. Recurrent autoimmune hepatitis negatively affects outcomes after liver transplantation. Thus, improved measures are required to prevent and treat this condition.

Risk factors and outcomes associated with recurrent autoimmune hepatitis following liver transplantation

Aldo J. Montano-Loza^{1,*}, Vincenzo Ronca², Maryam Ebadi¹, Bettina E. Hansen³,
Gideon Hirschfield³, Saleh Elwir⁴, Mohamad Alsaed⁴, Piotr Milkiewicz⁵, Maciej K. Janik⁵,
Hanns-Ulrich Marschall⁶, Maria Antonella Burza⁶, Cumali Efe⁷, Ali Rıza Calışkan⁸,
Murat Harputluoglu⁸, Gökhan Kabaçam⁹, Débora Terrabuio¹⁰, Fernanda de Quadros Onofrio³,
Nazia Selzner³, Alan Bonder¹¹, Albert Parés¹², Laura Llovet¹², Murat Akyıldız¹³,
Cigdem Arikan¹⁴, Michael P. Manns¹⁵, Richard Taubert¹⁵, Anna-Lena Weber¹⁵,
Thomas D. Schiano¹⁶, Brandy Haydel¹⁶, Piotr Czubkowski¹⁷, Piotr Socha¹⁷, Natalia Ołdak¹⁷,
Nobuhisa Akamatsu¹⁸, Atsushi Tanaka¹⁹, Cynthia Levy²⁰, Eric F. Martin²⁰, Aparna Goel²¹,
Mai Sedki²¹, Irena Jankowska²², Toru Ikegami²³, Maria Rodriguez²⁴, Martina Sterneck²⁴,
Christina Weiler-Normann²⁴, Christoph Schramm²⁴, Maria Francesca Donato²⁵,
Ansgar Lohse²⁶, Raul J. Andrade²⁷, Vilas R. Patwardhan¹¹, Bart van Hoek²⁸,
Maaike Biewenga²⁸, Andreas E. Kremer^{29,30}, Yoshihide Ueda³¹, Mark Deneau³²,
Maria Francesca Secchi³⁵, Benedetta Terziroli Beretta-Piccoli³⁶, Marco Sciveres³⁷,
Giuseppe Maggiore³⁸, Syed-Mohammed Jafri³⁹, Dominique Debray⁴⁰, Muriel Girard⁴⁰,
Florence Lacaille⁴¹, Ellina Lytvyak¹, Andrew L. Mason¹, Michael Heneghan⁴², Ye Htun Oo ^{43,*},

¹Division of Gastroenterology and Liver Unit, University of Alberta, Edmonton, AB, Canada; ²Center for Liver Research & NIHR Birmingham BRC. University of Birmingham & University Hospital Birmingham NHS Foundation Trust. Institute of Immunology and Immunotherapy. University of Birmingham, UK; ³Toronto Center for Liver Disease, University Health Network, University of Toronto, Toronto, Canada; ⁴Baylor University Medical Center, Dallas, USA; ⁵Liver and Internal Medicine Unit, Medical University of Warsaw, Poland; ⁶Sahlgrenska University Hospital, Gothenburg, Sweden; ⁷Department of Gastroenterology, Harran University Hospital, Sanhurfa, Turkey; ⁸Department of Gastroenterology, Inönü University School of Medicine, Malatya, Turkey; ⁹Clinic of Gastroenterology and Liver Transplantation, Guven Hospital Ankara, Turkey; ¹⁰Department of Gastroenterology - University of São Paulo School of Medicine, São Paulo, Brazil; ¹¹Beth Israel Deaconess Medical Center, Harvard Medical School, Boston, USA; ¹²Liver Unit, Hospital Clínic, University of Barcelona, IDIBAPS, CIBERehd, Barcelona, Spain; ¹³Koç University School of Medicine, Department of Gastroenterology and Liver Transplantation Center, Istanbul, Turkey; ¹⁴Koc University School of Medicine, Pediatric Gastroenterology and Hepatology, Organ Transplantation Center, Koc University Research Center for Translational Medicine (KUTTAM), Istanbul, Turkey;¹⁵European Reference Network on Hepatological Diseases (ERN RARE-LIVER), Department Gastroenterology, Hepatology and Endocrinology, Hannover Medical School, Hannover, Germany; ¹⁶Recanati/Miller Transplantation Institute/Division of Liver Diseases, Mount Sinai Medical Center, New York, USA; ¹⁷Department of Gastroenterology, Hepatology, Nutritional Disorders and Pediatrics, The Children's Memorial Health Institute, Warsaw, Poland; ¹⁸University of Tokyo, Japan; ¹⁹Department of Medicine, Teikyo University School of Medicine, Tokyo, Japan; ²⁰University of Miami Miller School of Medicine, Miami, USA;
 ²¹Stanford University, Stanford, USA; ²²Children's Memorial Health Institute, Warsaw, Poland; ²³Department of Surgery and Science, Graduate School of Medical Sciences, Kyushu University, Fukuoka, Japan; ²⁴UKE Hamburg, Hamburg, Germany; ²⁵Foundation IRCCS Ca' Granda Ospedale Maggiore Policlinico, Liver Tranplant Hepatology Unit, Division of Gastroenterology and Hepatology, Milan, Italy; ²⁶University Medical Center, Hamburg, Germany; ²⁷Gastroenterology Service -IBIMA. University Hospital and CIBERehd. University of Málaga, Spain; ²⁸Leiden University Medical Center, Leiden, Netherlands; ²⁹Department of Medicine, University Hospital Erlangen and Friedrich-Alexander-University Erlangen-Nürnberg, Erlangen, Germany; ³⁰Department of Gastroenterology and Hepatology, University Hospital Zürich, University of Zürich, Zürich, Switzerland; ³¹Department of Gastroenterology and Hepatology, Graduate School of Medicine, Kyoto University, Kyoto, Japan; ³²University of Utah, Salt Lake City, USA; ³³University of Texas Southwestern Medical Center, Dallas, USA; ³⁴Department of Surgery, Oncology and Gastroenterology, University of Padova, Padova, Italy; ³⁵University of Padova, Padova, Italy; ³⁶Epatocentro Ticino & Università della Svizzera Italiana, Lugano, Switzerland; ³⁷UPMC Pediatric Liver Center, Palermo, Italy; ³⁸Hepatogastroenterology, Nutrition and Liver Transplant IRCCS Bambino Gesù Pediatric Hospital, Rome, Italy; ³⁹Henry Ford Health System, Detroit, Michigan, USA; ⁴⁰Pediatric Liver Unit, French National Reference Center for Rare Diseases BA and Genetic Cholestasis, Hôpital Necker, Université de Paris, Paris, France;

Received 6 May 2021; received in revised form 26 January 2022; accepted 27 January 2022; available online 8 February 2022

* Corresponding authors. Addresses: Division of Gastroenterology and Liver Unit, 8540 112 Street NW, Zeidler Ledcor Center, University of Alberta, Edmonton, AB, T6G 2X8, Canada; Tel.: (780) 248-1892, fax: (780) 248-1895. (A.J. Montano-Loza), or Centre for Liver and Gastro Research, University of Birmingham, Liver Transplant and Hepatobiliary Unit, Centre for Rare Disease & ERN Rare Liver Centre, UHB NHS Foundation Trust, United Kingdom. (Y.H. Oo). *E-mail addresses:* montanol@ualberta.ca (A.J. Montano-Loza), y.h.oo@bham.ac.uk (Y.H. Oo).

https://doi.org/10.1016/j.jhep.2022.01.022





Keywords: autoimmune liver disease; recurrent disease; survival; graft survival; liver transplantation.

⁴¹Gastroenterology-Hepatology-Nutrition Unit, Hôpital Necker-Enfants malades, Paris, France; ⁴²King's College Hospital NHS Foundation Trust, London, UK; ⁴³Center for Liver and Gastro Research & National Institute of Health Research Birmingham Biomedical Research Centre, University of Birmingham, Centre for Rare Disease and ERN Rare Liver Centre, Liver Transplant and Hepatobiliary Unit, University Hospital Birmingham NHS Foundation Trust, UK

Background & Aims: Autoimmune hepatitis can recur after liver transplantation (LT), though the impact of recurrence on patient and graft survival has not been well characterized. We evaluated a large, international, multicenter cohort to identify the probability and risk factors associated with recurrent AIH and the association between recurrent disease and patient and graft survival.

Methods: We included 736 patients (77% female, mean age 42±1 years) with AIH who underwent LT from January 1987 through June 2020, among 33 centers in North America, South America, Europe and Asia. Clinical data before and after LT, biochemical data within the first 12 months after LT, and immunosuppression after LT were analyzed to identify patients at higher risk of AIH recurrence based on histological diagnosis.

Results: AIH recurred in 20% of patients after 5 years and 31% after 10 years. Age at LT \leq 42 years (hazard ratio [HR] 3.15; 95% CI 1.22-8.16; p = 0.02), use of mycophenolate mofetil post-LT (HR 3.06; 95% CI 1.39-6.73; p = 0.005), donor and recipient sex mismatch (HR 2.57; 95% CI 1.39-4.76; p = 0.003) and high IgG pre-LT (HR 1.04; 95% CI 1.01-1.06; p = 0.004) were associated with higher risk of AIH recurrence after adjusting for other confounders. In multivariate Cox regression, recurrent AIH (as a time-dependent covariate) was significantly associated with graft loss (HR 10.79, 95% CI 5.37-21.66, p < 0.001) and death (HR 2.53, 95% CI 1.48-4.33, p = 0.001).

Conclusion: Recurrence of AIH following transplant is frequent and is associated with younger age at LT, use of mycophenolate mofetil post-LT, sex mismatch and high IgG pre-LT. We demonstrate an association between disease recurrence and impaired graft and overall survival in patients with AIH, highlighting the importance of ongoing efforts to better characterize, prevent and treat recurrent AIH.

Lay summary: Recurrent autoimmune hepatitis following liver transplant is frequent and is associated with some recipient features and the type of immunosuppressive medications use. Recurrent autoimmune hepatitis negatively affects outcomes after liver transplantation. Thus, improved measures are required to prevent and treat this condition.

© 2022 European Association for the Study of the Liver. Published by Elsevier B.V. All rights reserved.

Introduction

Autoimmune hepatitis (AIH) is a multifaceted liver disease, characterized by high IgG, autoantibodies, interface hepatitis on histological examination, and in most cases, an appropriate response to immunosuppression.^{1,2}

Liver transplantation (LT) is a lifesaving intervention for patients with advanced disease, and 1- and 5-year survival rates are approximately 90% and 70%, respectively.^{3–5} Recurrent AlH (*r*AlH) is a major cause of allograft dysfunction, reduced graft and patient survival, and need for re-transplantation. The prevalence of *r*AlH ranges from 17-42%, and this divergence seems to be related to differences in studies with respect to the use of protocol *vs.* clinically indicated liver biopsies, the small number of patients in each series, and variable follow-up times.^{6–8} Previous studies have also suggested that the development of *r*AIH has no significant impact on long-term patient survival or the need for a second LT^{6,9} and thus its clinical significance has been questioned. However, these observations may be related to inadequate follow-up and limited numbers of patients. In fact, recent studies from the United Network for Organ Sharing (UNOS) and the European Liver Transplant Registry (ELTR) have demonstrated that overall survival after LT is inferior in AIH compared to patients receiving LT for the other chronic auto-immune liver diseases.¹⁰

The impact of immunosuppression after LT on *r*AlH risk is still controversial. Longstanding low-dose corticosteroid (predniso-lone 5-10 mg) in combination with other immunosuppressive agents seems to reduce *r*AlH without risking patient and graft survival⁹; however, the most recent guidance from the American Association for the Study of Liver Diseases (AASLD) recommends glucocorticoids can be discontinued after LT and patients monitored for *r*AlH.¹¹

In addition, the importance of alterations in serum liver tests shortly after LT, remains unknown in AIH, whereas early serum liver test abnormalities have been associated with recurrent disease and worse outcomes in patients with primary biliary cholangitis (PBC) and primary sclerosing cholangitis (PSC).¹²

Accordingly, we conducted a multicenter study in 33 LT centers to evaluate the probability and risk factors associated with *r*AIH and the association between *r*AIH and patient and graft survival. In addition, we evaluated whether serum liver tests within the first year after LT were associated with subsequent recurrent disease. We hypothesized that serum liver test abnormalities during the first year after LT increase the risk for *r*AIH which, in turn, negatively impacts graft and patient survival.

Patients and methods

Study population

Eight hundred and fifty-five patients who received a LT from 1987 until 2020 with a diagnosis of AIH from 33 centers across Asia, Europe, North and South America, were evaluated (Supplementary Document 1 and Fig. S1). We excluded 119 patients who had features of overlap syndrome with PBC or PSC. All patients included were considered to have probable or definitive AIH according to simplified criteria,¹³ and to the AASLD¹⁴ and European Association for the Study of the Liver (EASL) guide-lines.² Based on the median number of LTs performed by all centers, those contributing more than 16 LTs for AIH were designated as high-volume centers.

Clinical and laboratory assessments

The data extracted from the medical records included sex, ethnicity, age at diagnosis of AIH and LT, time between diagnosis of AIH and LT, blood group, concomitant autoimmune disease, antinuclear antibodies, smooth muscle antibodies, anti-liver kidney microsome type 1 antibodies, IgG concentrations before LT, treatment for AIH, LT indication (acute liver or decompensated cirrhosis), type of LT and biliary anastomosis, and model for end-stage liver disease

score.¹⁵ We also recorded initial immunosuppression post-LT, longterm prednisone or prednisolone use (more than 1- and 5-years), initial trough levels of tacrolimus or cyclosporine, changes in immunosuppression after the first year of LT, liver biopsies after LT (protocol or clinically driven), year of LT, and number of LTs per center. In addition, we recorded donor age and sex, donor/recipient sex mismatch, donor's blood group, explant necroinflammatory activity and fibrosis score and rejection episodes.

In order to minimize the risk of variation in data collection, we discussed this project in our bi-annual meetings of the International AIH study group and developed instructions to standardize the collection of data. Serum liver function tests and immunosuppression levels were assessed every 1 to 3 months in all centers.

Liver serum tests including alanine aminotransferase (ALT), aspartate aminotransferase (AST), alkaline phosphatase and bilirubin were collected at the time of LT, and 3-, 6-, and 12-months after LT. Values, both raw and divided by the upper limit of normal (ULN), based on center-specific values were taken into account for the analyses. The ULN for ALT ranged from 31 to 56 U/L, AST from 30 to 52 U/L, IgG from 16 to 18 g/L, and bilirubin from 18 to 22 µmol/L between the different LT centers.

Histological assessment of liver explants and liver biopsies after liver transplantation

Histological assessment of the liver explant and liver biopsies for *r*AlH diagnosis was graded according to the Batts–Ludwig scoring system.¹⁶ On the basis of the Batts–Ludwig scoring system, grade 1 denotes necroinflammatory activities largely confined to the portal areas, with grades 2–3 representing extension beyond the portal areas and grade 4 signifying confluent necrosis in the form of bridging necrosis. Grade 0 represents no fibrosis, grade 1 conveys portal fibrosis, grade 2 signifies periportal fibrosis, grade 3 denotes septal fibrosis, and grade 4 indicates cirrhosis.

Diagnosis of recurrent autoimmune hepatitis

The diagnosis of *r*AlH was made histologically and defined by the presence of liver histology typical of AlH, positive autoantibodies, high IgG and negative viral hepatitis tests. Histologic features of *r*AlH were the presence of lymphoplasmacytic portal inflammation with interface hepatitis, with or without the presence of rosettes of regenerating hepatocytes, and hepatocyte emperipolesis in the absence of allograft rejection findings, such as endothelialitis, lymphocytic cholangitis or perivenular hepatocyte necrosis.¹ The simplified AlH score¹³ was calculated in patients with a histological diagnosis of *r*AlH. In addition, the presence of other infections and concomitant use of potentially hepatotoxic drugs were ruled out.

Immunosuppressive regimens

The type of immunosuppression administered during the first year was recorded. The predominant calcineurin inhibitor (either cyclosporine or tacrolimus) and other immunosuppressive medications (including azathioprine, mycophenolate mofetil, and prednisone or prednisolone) were all recorded. Changes in the main immunosuppression after the first year of LT were also recorded.

Statistical analyses

The Fisher exact probability test was used to compare categorical variables, and the unpaired t test was used to compare differences in means of continuous variables. Prognostic factors for

*r*AIH were analyzed by Cox regression univariate analysis.¹⁷ Variables with *p* value ≤ 0.05 in the univariate analysis were included in the Cox regression multivariate analysis to avoid model overfitting. IgG and age cut-offs at diagnosis and at the time of LT were established based on median to reduce the risk of optimist bias. Cumulative incidences of *r*AIH after LT were calculated using the Kaplan-Meier method, and they were compared using the Log-Rank (Mantel-Cox) test.¹⁸

To determine whether *r*AIH was significantly associated with graft loss and overall survival, the impact of rAIH on the hazard ratio of graft loss and survival was assessed using univariate and multivariate Cox regression analyses. In these analyses, the time until patients had rAIH was modelled as a time-dependent covariate. The association of rAIH with graft loss and overall survival was also analyzed as a time-dependent covariate. Variables with $p \le 0.05$ in the univariate analysis and other relevant variables were included in the Cox proportional hazard regression multivariate analysis. Patients that did not develop rAIH and died and those who were lost during follow-up were censored at the time of death or at the time of their last visit. In order to analyze the clinical impact of *r*AIH, patients who died or lost the graft within the first 3 months after LT were excluded from the survival analysis, as these outcomes were deemed related to surgical complications. Graft loss was defined using a death-censored definition of graft failure and therefore, graft loss did not include patients who died with a functioning graft. Graft loss only included deaths secondary to or associated with graft failure (i.e. recurrent disease, de novo disease, chronic ductopenic rejection, sepsis in patients with biliary or vascular complications, or cirrhosis development on the graft) or re-transplantation.

Cumulative probabilities of graft and overall survival after LT were calculated using semi-Markov models (so-called "clock reset" models) because each time the patient enters a new state, time is reset to 0 (in this case *r*AIH).¹⁹ As the median survival time for our patient population was not reached, the mean 5-year, 10-year, 15-year and 20-year survival probabilities were reported.

In order to evaluate potential limitations of estimation of probabilities and impact of *r*AIH related to lack of liver biopsies in all patients, we performed subanalysis including only patients with liver biopsies after LT (n = 529). Data are presented as the mean \pm standard error in tables and text, and median with interquartile ranges (IQR) in case data was not normally distributed. Statistical analyses were conducted using SPSS 26.0. (supplementary CTAT table). To handle the missing variables in the analysis, we used mean imputation for continuous variables and allocated a fixed number (99) for categorical variables.

Results

Characteristics, frequency and probability of autoimmune hepatitis recurrence

The mean age of the study population at LT was 42±1 years (median, 43 years; IQR: 41-45 years), and 563 patients (77%) were women. The 736 patients analyzed were transplanted because of acute liver failure secondary to AIH (n = 136, 19%), decompensated cirrhosis (n = 560, 76%) and hepatocellular carcinoma (n = 40, 5%). The main features of patients who received a LT for AIH are shown in Table 1. Recurrent AIH was diagnosed in 147 patients that represented 20% of the study population (n = 736) and 28% of those who had a liver biopsy after LT (n = 529). According to the simplified AIH score at the time of recurrence,¹³ 38 patients (26%) were classified as likely AIH (\geq 7 points), 93

Table 1. Clinical feat	ures associated with	recurrent AIH in	univariable analyses.
------------------------	----------------------	------------------	-----------------------

Age at lume of chapnois All (years) 3441 7271 3021 038 007.999 0000 Age at dignois St (years), (%) 306 (42) 82 (56) 224 (38) 1.76 $123.25.8$ 0004 Age at C (years), (%) 350 (48) 99 (07) $123.42.25$ 0004 Men: women 173.2563 28.119 144.44 0.75 $0.39.116$ 0.38 Biod group (recipient) 71.205 82.019 72.204 0.99 $0.97.102$ 0.03 Biod group (recipient) 71.205 $0.52.119$ 0.444 0.5 $0.7.189$ 0.83 O concentiant autoimmure disease. n (X) 153 (21) 0.12 $0.7.489$ $0.027.499$ $0.027.4$	Clinical features	All patients (n = 736)	<i>r</i> AIH (n = 147)	No <i>r</i> AIH (n = 589)	HR	95% CI	p value
Age at digenois :34 (vers), n (3) 306 (42) 82, (56) 224 (38) 1.75 120-2.58 0.004 Age at L (yers), n (3) 350 (48) 99 (7) 251 (43) 1.91 1.91 - 2.92 4.001 Mer: worner 173 556 (28) 281 (19) 1454.44 0.05 0.43 + 11 0.18 Mer: worner 173 556 (28) 281 (19) 145.444 0.05 0.43 + 11 0.18 Mood group (repiers) 71403 526 (19) 192 (17) 29 (17) - - - 0.03 0.13 + 114 0.04 0.03 + 114 0.05 0.73 + 140 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 114 0.04 0.03 + 118 0.08 0.03 + 114 0.04 + 135 0.07 + 130 0.07 + 130 0.04 + 130 0.04 + 130 0.04 + 130 0.04 + 130 0.04 + 130 0.04 + 130 <t< td=""><td>Age at the time of diagnosis AIH (years)</td><td>34±1</td><td>27±1</td><td>36±1</td><td>0.98</td><td>0.97-0.99</td><td>< 0.001</td></t<>	Age at the time of diagnosis AIH (years)	34±1	27±1	36±1	0.98	0.97-0.99	< 0.001
Age at IT solvers)42zi13zi44zi0.980.97-0.99et.0001Met: women17355328:1191454440.731511.32-220.001Met: women29134576:71315:240.330.59-1160.25Encusian: non-Caucaian29134576:71315:240.330.59-1160.25Encusian: non-Caucaian391:34576:71315:240.330.59-1160.25A262(36)9601729(5)777	Age at diagnosis ≤34 (years), n (%)	306 (42)	82 (56)	224 (38)	1.76	1.20-2.58	0.004
Age at IF 42 (years), n (3) 350 (48) 99 (67) 251 (43) 191 (43, 42, 40) 134-272 0.001 Men: women 737, 556 28,119 145,244 0.97 0.93 (.97) 0.26 Time Att diagnosis and IT (years) 71,03 58,050 722,04 0.99 0.97-1,160 0.26 Rod 29 (.5) 724,04 0.99 0.97-1,160 0.26 A 29 (.5) 29 (.5) 0.73 84 (.14) 0.45 0.05 0.47-1,480 0.85 0 341 (46) 81 (55) 226 (.40) 0.95 0.47-1,89 0.85 0 341 (46) 81 (55) 226 (.40) 0.55 0.47-1,89 0.85 0 341 (46) 81 (55) 226 (.40) 1.04 0.55-1,83 0.02 Instancempression before IT, n (3): T 1.04 (.21,2 0.75-75 0.1 Acatanycin 0.05 1.47 (.43) 0.24 (.44) 1.04 0.55-1,453 0.03 Upcompressicat chrobis (incluing HCC) 600 (.82) 124 (.44)	Age at LT (years)	42±1	35±1	44±1	0.98	0.97-0.99	< 0.001
Men: women 173:563 28:119 143:444 0.75 0.49-1.15 0.08 Carculars: non-Carcularia" 315:274 0.83 0.59-1.16 0.26 Biod group (recipient) 226:213 10.(7) 29.15 0.89 0.97-1.02 0.60 Biod group (recipient) 226:213 10.(7) 29.15 0.33-1.64 0.55 0.55 0.07-1.25 0.00 Concombrant attributing mode disease. n (3) 153 (21) 40.027 113 (19) 1.56 1.07-2.26 0.04 Immunosuppression before LF, n (3): Prednisone 501 (168) 0.03 (70) 398 (68) 1.12 0.78-1.62 0.04 Austhoprine 314 (144) 74 (50) 267 (45) 1.13 0.03-1.198 0.08 Mycophenolate modeil 00 (8) 11 (8) 224 (18) 110 (19) 0.24-1.88 0.002 Indicatorin, non-Caracutorino function precision post-LT, n (%): Transmither 117 117.3-1.73 0.001 Tindication, n (%): Transmither 516 (70) 127 (11) 429 (76)	Age at LT ≤42 (years), n (%)	350 (48)	99 (67)	251 (43)	1.91	1.34-2.72	<0.001
Caucasian: non-Caucasian' 391:345 78:71 315:274 0.83 0.59-1.16 0.26 Time Alth digamos and I' (years) A 30 (5 10 (7 29 (5) A 202 (3) 446 (31) 216 (37) 0.80 0.39-1.64 0.55 A 202 (3) 446 (31) 216 (37) 0.80 0.39-1.64 0.55 B 0 documental autoimmune disease. n (8) 131 (21) 40 (27) 113 (19) 1.50 107-226 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-226 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-226 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-226 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-226 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-256 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-256 0.02 Immediate autoimmune disease. n (8) 133 (21) 40 (27) 113 (19) 1.50 107-256 0.02 IT indication. n (8) Acute liver failare 16 (18) 22 (16) 113 (19) 1. Decompensated circlosis (including HCC) 600 (82) 124 (84) 49 (8) 104 0.54-189 0.48 Type of II, n (8) Acute liver failare 134 (19) 11 (7) 132 (22) 0.35 0.16-068 0.002 IT indication. n (8) Acute liver failare 143 (19) 11 (7) 132 (22) 0.35 0.16-068 0.002 It indicat autoimmus, n (8) Acute liver failare 143 (19) 11 (7) 132 (22) 0.37 0.51-106 0.09 It indicat autoimmus, n (8): Eact-0-eant 9 Bind durat autoimmus, n (8): Eact-0-eant 9 Bind durat autoimmus, n (8): It indicat autoimmus, n (8): It is apprediate 14, n (8): It is a set of the first year 16 (23) (24) (14) (24) (25) 0.004 Bind durat autoimmus, n (8): It is a set of the first year 17 (23) (23) (24) (24) (25) 0.004 It is a set of the first year 17 (23) (23) (24) (24) (25) 0.046 Bind durat year 17 (23) (24) (24) (25) 0.046 0.053 It is a maximouppression after the first year of 17 96 (1) 13 (9) (21) (14) 280 0.045-106 Door 29 (0000) 0.22	Men: women	173:563	28:119	145:444	0.75	0.49-1.15	0.18
Time AH diagnosis and IT (years) 7.14.0. 6.840.6 7.24.0. 0.99 0.97-1.02 0.60 Bod group (recipien) 39 (5) 10 (7) 29 (5)	Caucasian: non-Caucasian [¥]	391:345	76:71	315:274	0.83	0.59-1.16	0.26
Biol Biol Biol Display Display <thdisplay< th=""> <thdisplay< th=""> <thdisplay< t<="" td=""><td>Time AIH diagnosis and LT (years)</td><td>7.1±0.3</td><td>6.8±0.6</td><td>7.2±0.4</td><td>0.99</td><td>0.97-1.02</td><td>0.60</td></thdisplay<></thdisplay<></thdisplay<>	Time AIH diagnosis and LT (years)	7.1±0.3	6.8±0.6	7.2±0.4	0.99	0.97-1.02	0.60
A 29 (5) 10 (7) 29 (5) A 262 (36) 40 (37) 0.080 0.39-1.64 0.55 B 94 (13) 10 (7) 284 (14) 0.055 0.47-1.89 0.88 Concomitant autoimmune disease, n (3) 133 (21) 40 (27) 113 (19) 1.56 1.07-2.26 0.02 Immunosuppression before LT, n (3): """"""""""""""""""""""""""""""""""""	Blood group (recipient)						
A 262 (36) 46 (31) 216 (37) 0.00 0.39-1.64 0.05 B 94 (13) 10 (7) 84 (14) 0.45 0.47-1.89 0.88 Concomitant autoimmune disease, n (3) 153 (21) 402 (27) 113 (19) 1.56 1.07-2.26 0.02 Immunosuppression before II, n (3): " " 144 (46) 74 (50) 257 (43) 1.35 0.03 Acathioprine 31 (46) 74 (50) 257 (43) 1.35 0.03 0.78-5.75 0.1 Acathioprine 31 (46) 74 (50) 257 (14) 0.35 0.03 0.35 0.03 0.02 Indication, f(3) " " " " " 0.02 0.14 (44) 76 (81) 1.19 0.74-1.89 0.48 Up of L1, n (5): " " " " " " 0.02 0.02 0.03 0.08-08 0.002 Itidi intramonsuppression before L1, n (3): " 1.14 (14) " 0.47 0.45 0.002 0.14 0.63 (11) 0.46 0.49.09 0.05 0.02 <td< td=""><td>AB</td><td>39 (5)</td><td>10 (7)</td><td>29 (5)</td><td></td><td></td><td></td></td<>	AB	39 (5)	10 (7)	29 (5)			
B 94 (13) 10 (7) 84 (14) 0.05 0.47-1.89 0.88 Concomitant autoimmume disease, n (3) 153 (21) 40 (27) 113 (19) 1.56 1.07-2.26 0.02 Immunosuppression hefore LT, n (3): """"""""""""""""""""""""""""""""""""	Α	262 (36)	46 (31)	216 (37)	0.80	0.39-1.64	0.55
0 341 (46) 81 (55) 250 (44) 0.02 0.47-1.89 0.88 Concomitant automune disease, n(X) 153 (21) 401 (27) 113 (19) 1.15 0.02 Immunosuppression before IT, n(X): Prednisone 101 (3) 5 (4) 14 (2) 2.12 0.78-5.75 0.11 Acathoprine 311 (46) 74 (50) 257 (45) 1.35 0.078-5.75 0.11 Acathoprine 311 (46) 74 (50) 257 (48) 1.35 0.078-5.75 0.11 Acathoprine 311 (46) 74 (50) 257 (48) 1.35 0.027-1.89 0.08 Decompensate infrois (induding HCC) 008 (22) 124 (44) 476 (81) 1.19 0.74-1.89 0.48 Dig effat 113 (19) 117 (17) 112 (12) 0.08 0.002 Ed dust ausomasis, n (%): = = 1.17-3.17 0.009 Indications or predisions opst-LT, n (%): 557 (78) 91 (33) 452 (32) 0.04 0.07 Indicatins or predinisolone, n (%): =	В	94 (13)	10(7)	84 (14)	0.45	0.18-1.11	0.08
Conconsistant autoimmune disease, n (3) (3) (3) (3) (70) (38 (68) (12) (0.7-2.26 (0.02) (3) (70) (38 (68) (12) (0.7-2.26 (0.02) (3) (70) (30) (70) (70) (70) (70) (70) (70) (70) (7	0	341 (46)	81 (55)	260 (44)	0.95	0.47-1.89	0.88
Immunosuppresion before LT, n (%): Second (68) 010 (70) 398 (88) 1.12 0.78-1.62 0.54 Budesonide 19 (3) 5 (4) 14 (2) 2.12 0.78-5.75 0.11 Actuitoprine 31 (46) 74 (50) 2.67 (45) 1.35 0.97-1.89 0.08 Mycophenolate moletil 60 (8) 11 (8) 49 (8) 1.04 0.54-1.98 0.02 Cataveric 136 (18) 23 (16) 113 (19) - - - - Decompensated cirrhosis (including HCC) 600 (82) 12.4 (40) 476 (8) 1.09 0.74-1.89 0.04 -	Concomitant autoimmune disease, n (%)	153 (21)	40 (27)	113 (19)	1.56	1.07-2.26	0.02
Predinsone 501 (68) 103 (70) 398 (68) 1.12 0.78-5.75 0.11 Azathoprine 341 (46) 74 (50) 267 (45) 1.35 0.078-5.75 0.11 Azathoprine 341 (46) 74 (50) 267 (45) 1.35 0.078-5.75 0.11 Azatte liver failure 136 (18) 221 (6) 113 (19) -	Immunosuppression before LT, n (%):						
Budesonide 19 (3) 5 (4) 14 (2) 2.12 0.78.57.5 0.11 Azathioprine 314 (4) 74 (6) 2.67 (45) 1.35 0.097.18.9 0.08 Mycophenolate mofeli 60 (8) 11 (8) 49 (8) 1.04 0.54-1.98 0.028 Decompensate drinkos (including HCC) 136 (18) 2.32 (16) 113 (19) 0.74-1.89 0.48 Type of L1, n (%): Calaveric 516 (70) 12.4 (84) 476 (8) 1.09 0.74-1.89 0.04 Bie duct anastomesis, n (%): T Tass (76) 193 (17) 449 (76) T Tomosis 0.02 0.024 (14) 0.017 1.93 0.035 0.038-0.68 0.002 Bie duct anastomesis, n (%): T Tass (76) 193 (13) 449 (76) T Tomosis 0.05 7.75 (76) 101 (169) 456 (18) 0.07 0.049.0.9 0.05 7.75 (76) 101 (169) 456 (18) 0.07 0.05 (11.0.9 0.05 (11.0.9 0.05 (11.0.9 0.05 (11.0.9 0.05 (11.0.9 0.05 (11.0.9 <td>Prednisone</td> <td>501 (68)</td> <td>103 (70)</td> <td>398 (68)</td> <td>1.12</td> <td>0.78-1.62</td> <td>0.54</td>	Prednisone	501 (68)	103 (70)	398 (68)	1.12	0.78-1.62	0.54
Azathoprine 341 (46) 74 (50) 267 (45) 1.35 0.97-1.89 0.08 Mycophenolace moteril 060 (8) 11 (8) 49 (8) 1.04 0.54-1.98 0.92 LT indication, n (3) Acate liver failure 136 (18) 23 (16) 113 (19)	Budesonide	19 (3)	5 (4)	14 (2)	2.12	0.78-5.75	0.1
Mycophenolate motenil 60 (8) 11 (8) 49 (8) 1.04 0.54-1.98 0.92 Acate liver failure 136 (18) 23 (16) 113 (19)	Azathioprine	341 (46)	74 (50)	267 (45)	1.35	0.97-1.89	0.08
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Mycophenolate mofetil	60 (8)	11 (8)	49 (8)	1.04	0.54-1.98	0.92
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	LT indication, n (%)						
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Acute liver failure	136 (18)	23 (16)	113 (19)			
Type of L n (%): 516 (70) 124 (84) 392 (67) Living related 143 (19) 11 (7) 132 (22) 0.35 0.18-0.68 0.002 Bile dust anaxomosis, n (%): 554 (75) 105 (71) 449 (76) 70009 Roux-en-Y 60 (8) 20 (14) 449 (76) 70009 70009 Initial immunosuppression post-LT, n (%): Tarcolinuus 575 (78) 93 (63) 452 (82) 0.70 0.49-0.99 0.05 Cyclosporine 80 (11) 32 (22) 48 (8) 167 1.09-2.56 0.002 Mycophenolate Mofetil 404 (55) 73 (50) 331 (56) 159 111-2.27 0.01 Azathoprine 76 (10) 13 (9) 63 (11) 0.46 0.26-0.82 0.008 Strolimus 15 (2) 1 (1) 14 (2) 0.47 0.65-3.37 0.44 > 5 years 362 (49) 87 (59) 275 (47) 115 0.82-162 0.41 > 5 years 15 (23) 5 (37) 115 (20) 114 0.80-161 </td <td>Decompensated cirrhosis (including HCC)</td> <td>600 (82)</td> <td>124 (84)</td> <td>476 (81)</td> <td>1.19</td> <td>0.74-1.89</td> <td>0.48</td>	Decompensated cirrhosis (including HCC)	600 (82)	124 (84)	476 (81)	1.19	0.74-1.89	0.48
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Type of LT. n (%):						
Living related 143 (19) 11 (7) 132 (22) 0.35 0.18-0.68 0.002 Bile duct anastomosis, n (%):	Cadaveric	516 (70)	124 (84)	392 (67)			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Living related	143 (19)	11 (7)	132 (22)	0.35	0.18-0.68	0.002
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	Bile duct anastomosis n (%):		(-)	()			
Rousen-Y 60 (8) 20 (14) 40 (7) 1.93 1.17-3.17 0.009 Initial immunosupression post-LT, n (%):	End-to-end	554 (75)	105 (71)	449 (76)			
$\begin{array}{ l l l l l l l l l l l l l l l l l l $	Roux-en-Y	60 (8)	20 (14)	40 (7)	1.93	1.17-3.17	0.009
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Initial immunosuppression post-LT, n (%):		. ,				
	Tacrolimus	575 (78)	93 (63)	482 (82)	0.70	0.49-0.99	0.05
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Cyclosporine	80 (11)	32 (22)	48 (8)	1.67	1.09-2.56	0.02
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Prednisone or prednisolone	557 (76)	101 (69)	456 (78)	0.73	0.51-1.05	0.09
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Mycophenolate Mofetil	404 (55)	73 (50)	331 (56)	1.59	1.11-2.27	0.01
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Azathioprine	76 (10)	13 (9)	63 (11)	0.46	0.26-0.82	0.008
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Sirolimus	15 (2)	1(1)	14 (2)	0.47	0.66-3.37	0.45
> i year362 (49)87 (59)275 (47)1.150.82-1.620.41> 5 years170 (23)55 (37)115 (20)1.140.80-1.610.48Tacrolimus (initial trough levels, ng/ml)82540.29.2620.58.440.21.020.98-1.010.32Cyclosporine levels (Initial trough levels, ng/ml)32742299±62344±560.990.99-1.0010.32Liver biopsies after L1, n (%):	Long-term prednisone-prednisolone, n (%)	- ()	()	()			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	>1 year	362 (49)	87 (59)	275 (47)	1.15	0.82-1.62	0.41
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	>5 years	170 (23)	55 (37)	115 (20)	1.14	0.80-1.61	0.48
Cyclosporine levels (Initial trough levels, ng/ml) 327±42 299±62 344±56 0.99 0.99-1.001 0.32 Liver biopsies after UT, n (%):	Tacrolimus (initial trough levels, ng/ml)	8.5±0.2	9.26±0.5	8.4±0.2	1.02	0.98-1.07	0.35
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	Cyclosporine levels (Initial trough levels, ng/ml)	327±42	299±62	344±56	0.99	0.99-1.001	0.32
Protocol204 (28)27 (18)177 (30)2.29Clinically driven325 (44)113 (77)212 (36)Changes in immunosuppression after the first year of LT56 (8)13 (9)43 (7)0.860.48-1.570.63LT center volume (High*: Low), n578:158117:30461:1281.080.72-1.610.72LT calendar year2010±0.32006±0.62010±0.30.990.96-1.010.28LT decade	Liver biopsies after LT, n (%):					1.50-3.50	< 0.001
Clinically driven325 (44)113 (77)212 (36)Changes in immunosuppression after the first year of LT56 (8)13 (9)43 (7)0.860.488-1.570.63LT center volume (High*: Low), n578:158117:30461:1281.080.72-1.610.72LT calendar year2010-0320064.6620104-030.990.96-1.010.28LT decade13 (9)21 (4)1.250.69-2.300.641998-2008215 (30)74 (50)141 (24)1.250.69-2.300.642008-2020460 (65)52 (35)448 (69)0.860.45-1.660.66Donor age (years)42:139±242:10.9960.99-1.010.50Donor sex (Men: women), n312:29664:57248:2391.110.77-1.610.57Donor/recipient sex mismatch, n(%)295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)42 (3)17 (3)4310.330.29-2.320.72A20 (30)37 (25)183 (31)0.830.29-2.320.720.32O422 (57)97 (66)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)TT1.321.320.240.84-1.850.28Explant fibrosis stage, n (%)T32 (63)329 (56)32 (55)0.970.36-2.650.96Explant fibrosis stage, n (%)S34 (6)3 (2)3	Protocol	204 (28)	27 (18)	177 (30)	2.29		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Clinically driven	325 (44)	113 (77)	212 (36)			
LT center volume (High*: Low), n 578:158 117:30 461:128 1.08 0.72-1.61 0.72 LT celendar year 2010±0.3 2006±0.6 2010±0.3 0.99 0.96-1.01 0.28 LT decade	Changes in immunosuppression after the first year of LT	56 (8)	13 (9)	43 (7)	0.86	0.48-1.57	0.63
LT calendar year 2010±0.3 2006±0.6 2010±0.3 0.99 0.96-1.01 0.28 LT decade	LT center volume (High*: Low), n	578:158	117:30	461:128	1.08	0.72-1.61	0.72
LT decade 1987-1997 34 (5) 13 (9) 21 (4) 1998-2008 215 (3) 74 (5) 141 (24) 1.25 0.69-2.30 0.66 2008-2020 460 (65) 52 (35) 408 (69) 0.86 0.45-1.66 0.66 Donor age (years) 42±1 39±2 42±1 0.996 0.99-1.01 0.50 Donor sex (Men: women), n 312:296 64:57 248:239 1.11 0.77-1.61 0.57 Donor frecipient sex mismatch, n (%) 295 (40) 70 (48) 225 (38) 1.48 1.02-2.15 0.04 Blood group (donor) -	LT calendar year	2010±0.3	2006±0.6	2010±0.3	0.99	0.96-1.01	0.28
1987-199734 (5)13 (9)21 (4)1988-2008215 (30)74 (50)141 (24)1.250.69-2.300.462008-2020460 (65)52 (35)408 (69)0.860.45-1.660.66Donor age (years)42±139±242±10.9960.99-1.010.50Donor sex (Men: women), n312:29664:57248:2391.110.77-1.610.57Donor vex (Men: women), n295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)42134 (3)17 (3)770.320.72A220 (30)37 (25)183 (31)0.830.29-2.320.720.32B21 (3)4 (3)17 (3)70.320.720.32O422 (57)97 (66)325 (55)0.970.63-6.650.920422 (57)97 (66)325 (55)0.970.63-6.650.92Crade 0/1/2392 (64)63 (43)329 (56)70.82Grade 3/4218 (36)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)531 (2)31 (5)51.240.25-6.140.79Stage 134 (6)3 (2)25 (4)1.240.25-6.140.7953.623.622.930.24Stage 3409 (6)38 (6)2.190.59-8.090.243.623.633.602.190.59-8.090.24<	LT decade						
1998-2008215 (30)74 (50)141 (24)1.250.69-2.300.462008-2020460 (65)52 (35)408 (69)0.860.45-1.660.66Donor age (years)42±139±242±10.9960.99-1.010.50Donor sex (Men: women), n312:29664:57248:2391.110.77-1.610.77Donor/recipient sex mismatch, n (%)295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)A21 (3)4 (3)17 (3)A220 (30)37 (25)183 (31)0.830.29-2.320.72B21 (3)4 (3)17 (3)A220 (30)37 (25)183 (31)0.830.29-2.320.72 <td< td=""><td>1987-1997</td><td>34 (5)</td><td>13 (9)</td><td>21 (4)</td><td></td><td></td><td></td></td<>	1987-1997	34 (5)	13 (9)	21 (4)			
2008-2020460 (65)52 (35)408 (69)0.860.45-1.660.66Donor age (years)42±139±242±10.9960.99-1.010.50Donor sex (Men: women), n312:29664:57248:2391.110.77-1.610.57Donor/recipient sex mismatch, n (%)295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)71 (3)4 (3)177 (3)737373 (10)9 (6)64 (11)0.550.17-1.790.32A220 (30)37 (25)183 (31)0.830.29-2.320.720220.720220.7200220.72000.660.660.17-1.790.320.72000.660.670.790.660.660.660.670.790.560.660.660.210.59-6.14<	1998-2008	215 (30)	74 (50)	141 (24)	1.25	0.69-2.30	0.46
Donor age (years)42±139±242±10.9960.99-1.010.50Donor sex (Men: women), n312:29664:57248:2391.110.77-1.610.57Donor/recipient sex mismatch, n (%)295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)AB21 (3)4 (3)17 (3)A220 (30)37 (25)183 (31)0.830.29-2.320.720.72B73 (10)9 (6)64 (11)0.550.17-1.790.32O422 (57)97 (66)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)Grade 0/1/2392 (64)63 (43)329 (56)Grade 3/4218 (36)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)Stage 134 (6)3 (2)31 (5)Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	2008-2020	460 (65)	52 (35)	408 (69)	0.86	0.45-1.66	0.66
Donor sex (Men: women), n312:29664:57248:2391.110.77-1.610.57Donor/recipient sex mismatch, n (%)295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)A21 (3)4 (3)17 (3)777 <t< td=""><td>Donor age (years)</td><td>42±1</td><td>39±2</td><td>42±1</td><td>0.996</td><td>0.99-1.01</td><td>0.50</td></t<>	Donor age (years)	42±1	39±2	42±1	0.996	0.99-1.01	0.50
Donor/recipient sex mismatch, n (%)295 (40)70 (48)225 (38)1.481.02-2.150.04Blood group (donor)AB21 (3)4 (3)17 (3)A220 (30)37 (25)183 (31)0.830.29-2.320.72B73 (10)9 (6)64 (11)0.550.17-1.790.32O422 (57)97 (66)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)73 (10)9 (6)63 (43)329 (56)73Grade 0/1/2392 (64)63 (43)329 (56)730.28Explant fibrosis stage, n (%)34 (6)3 (2)31 (5)31 (5)Stage 134 (6)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Donor sex (Men: women), n	312:296	64:57	248:239	1.11	0.77-1.61	0.57
Blood group (donor) AB 21 (3) 4 (3) 17 (3) A 220 (30) 37 (25) 183 (31) 0.83 0.29-2.32 0.72 B 73 (10) 9 (6) 64 (11) 0.55 0.17-1.79 0.32 O 422 (57) 97 (66) 325 (55) 0.97 0.36-2.65 0.96 Explant necroinflammatory activity score, n (%) 73 (10) 9 (6) 63 (43) 329 (56) 73 0.29 0.24 0.84-1.85 0.29 0.26 0.96 Grade 0/1/2 392 (64) 45 (31) 173 (29) 1.24 0.84-1.85 0.28 0.28 0.28 0.29 1.24 0.84-1.85 0.28 0.29 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.25 0.25 0.25 0.25 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.24 0.25 0.25 0.25 0.25 0.25 0.	Donor/recipient sex mismatch, n (%)	295 (40)	70 (48)	225 (38)	1.48	1.02-2.15	0.04
AB21 (3)4 (3)17 (3)A220 (30)37 (25)183 (31)0.830.29-2.320.72B73 (10)9 (6)64 (11)0.550.17-1.790.32O422 (57)97 (66)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)73 (20)392 (64)63 (43)329 (56)1.240.84-1.850.28Grade 0/1/2392 (36)45 (31)173 (29)1.240.84-1.850.280.28Explant fibrosis stage, n (%)34 (6)3 (2)31 (5)1.240.25-6.140.79Stage 134 (6)3 (2)25 (4)1.240.25-6.140.79Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Blood group (donor)						
A220 (30)37 (25)183 (31)0.830.29-2.320.72B73 (10)9 (6)64 (11)0.550.17-1.790.32O422 (57)97 (66)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)73 (20)173 (29)1.240.84-1.850.28Grade 0/1/2392 (64)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)54 (6)3 (2)31 (5)1515Stage 134 (6)3 (2)25 (4)1.240.25-6.140.79Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	AB	21 (3)	4 (3)	17 (3)			
B73 (10)9 (6)64 (11)0.550.17-1.790.32O422 (57)97 (60)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)392 (64)63 (43)329 (56)5050Grade 0/1/2392 (64)63 (43)329 (56)50505050Grade 3/4218 (36)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)5134 (6)3 (2)31 (5)5151Stage 134 (6)3 (2)25 (4)1.240.25-6.140.79Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	A	220 (30)	37 (25)	183 (31)	0.83	0.29-2.32	0.72
0422 (57)97 (66)325 (55)0.970.36-2.650.96Explant necroinflammatory activity score, n (%)Grade 0/1/2392 (64)63 (43)329 (56)Grade 3/4218 (36)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)Stage 134 (6)3 (2)31 (5)5151Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	В	73 (10)	9 (6)	64 (11)	0.55	0.17-1.79	0.32
Explant necroinflammatory activity score, n (%) Grade 0/1/2 392 (64) 63 (43) 329 (56) Grade 3/4 218 (36) 45 (31) 173 (29) 1.24 0.84-1.85 0.28 Explant fibrosis stage, n (%) 5tage 1 34 (6) 3 (2) 31 (5) 5 Stage 2 28 (5) 3 (2) 25 (4) 1.24 0.25-6.14 0.79 Stage 3 47 (8) 9 (6) 38 (6) 2.19 0.59-8.09 0.24 Stage 4 501 (82) 93 (63) 408 (69) 2.05 0.65-6.49 0.22	0	422 (57)	97 (66)	325 (55)	0.97	0.36-2.65	0.96
Grade 0/1/2392 (64)63 (43)329 (56)Grade 3/4218 (36)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)Stage 134 (6)3 (2)31 (5)Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Explant necroinflammatory activity score, n (%)						
Grade 3/4218 (36)45 (31)173 (29)1.240.84-1.850.28Explant fibrosis stage, n (%)Stage 134 (6)3 (2)31 (5)Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Grade 0/1/2	392 (64)	63 (43)	329 (56)			
Explant fibrosis stage, n (%)Stage 134 (6)3 (2)31 (5)Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Grade 3/4	218 (36)	45 (31)	173 (29)	1.24	0.84-1.85	0.28
Stage 134 (6)3 (2)31 (5)Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Explant fibrosis stage, n (%)						
Stage 228 (5)3 (2)25 (4)1.240.25-6.140.79Stage 347 (8)9 (6)38 (6)2.190.59-8.090.24Stage 4501 (82)93 (63)408 (69)2.050.65-6.490.22	Stage 1	34 (6)	3 (2)	31 (5)			
Stage 3 47 (8) 9 (6) 38 (6) 2.19 0.59-8.09 0.24 Stage 4 501 (82) 93 (63) 408 (69) 2.05 0.65-6.49 0.22	Stage 2	28 (5)	3 (2)	25 (4)	1.24	0.25-6.14	0.79
Stage 4 501 (82) 93 (63) 408 (69) 2.05 0.65-6.49 0.22	Stage 3	47 (8)	9 (6)	38 (6)	2.19	0.59-8.09	0.24
	Stage 4	501 (82)	93 (63)	408 (69)	2.05	0.65-6.49	0.22

(continued on next page)

Table 1. (continued)

All patients (n = 736)	<i>r</i> AIH (n = 147)	No <i>r</i> AIH (n = 589)	HR	95% CI	p value
297 (49)	58 (39)	239 (41)	1.12	0.75-1.66	0.58
254 (35)	82 (56)	172 (29)	2.04	1.45-2.85	<0.001
7 (1)	2 (1)	5(1)	0.93	0.23-3.77	0.92
	All patients (n = 736) 297 (49) 254 (35) 7 (1)	All patients rAlH (n = 736) (n = 147) 297 (49) 58 (39) 254 (35) 82 (56) 7 (1) 2 (1)	All patientsrAlHNo rAlH $(n = 736)$ $(n = 147)$ $(n = 589)$ 297 (49)58 (39)239 (41)254 (35)82 (56)172 (29)7 (1)2 (1)5 (1)	All patients rAlH No rAlH HR (n = 736) (n = 147) (n = 589) 112 297 (49) 58 (39) 239 (41) 1.12 254 (35) 82 (56) 172 (29) 2.04 7 (1) 2 (1) 5 (1) 0.93	All patients rAlH No rAlH HR 95% Cl (n = 736) (n = 147) (n = 589)

Hazard ratios were calculated using Cox proportional regression analyses. AIH, autoimmune hepatitis; HCC, hepatocellular carcinoma; LT, liver transplant; rAIH, recurrent AIH. *>17 LT performed for AIH.

[¥]Non-Caucasian includes, 8.0% Asian, 1.1% Aboriginal, 0.3% Pacific Islander, 5.0% Middle Eastern/Arabian, 4.5% Black, 8.7% Turkish, 19.3% Other/Multiracial.

patients (63%) were classified as probable AIH (6 points), and 16 patients (11%) as possible AIH (≤ 5 points). Once rAIH was diagnosed, prednisone or prednisolone was added in 61 (42%) patients, azathioprine was added in 19 (13%), prednisone or prednisolone and azathioprine was added in 20 (14%), mycophenolate mofetil was added in 14 (10%), prednisone or prednisolone and mycophenolate mofetil was added in 17 (12%), and only the baseline immunosuppression (either with tacrolimus or cyclosporine) was increased in 16 (11%). The median time from LT to the recurrence of AIH was 2.6 years (IQR: 1.2-5.7). The probability of rAIH was 20%, 31%, 37%, and 49% at 5-, 10-, 15-, and 20-years, respectively (Fig. 1). In addition, the probability of rAIH in patients with liver biopsies after LT (n = 529) was 24%, 38%, 44%, and 55% at 5-, 10-, 15-, and 20-years, respectively. The frequency of rAIH varied between 0% to 69% among centers. The yearly recurrence rate ranged from 0% to 10.1%. The overall incidence rate of rAIH after LT was 3.27 cases per 100 patientvears (95% CI 3.05–3.49 cases per 100 patient-vears over a total of 4,491 patient-years). In addition, the incidence rate of rAIH after LT, including only patients who had a liver biopsy after LT (n = 529) was 3.93 cases per 100 patient-years (95% CI 3.76-4.10 cases per 100 patient-years over a total of 3,564 patient-years).

The biochemical features after LT in patients with and without *r*AlH are presented in Table 2. The histological assessment of the liver explant was available for 610 patients (83%). Liver necroinflammatory activity was grade 0 in 47 patients (8%), grade 1 in 174 patients (29%), grade 2 in 171 patients (28%), grade 3 in 115 patients (19%) and grade 4 in 103 patients (17%). The fibrosis stage at the explant was stage 1 in 34 patients (6%), stage

2 in 28 patients (5%), stage 3 in 47 patients (8%), and stage 4 in 501 (82%). Two hundred and ninety-seven (49%) showed infiltration with plasma cells.

Biopsies after LT were performed in 529 patients (72%). Of those, 204 patients (39%) underwent protocol liver biopsies whereas 325 patients (61%) had clinically driven biopsies with abnormal serum liver tests. The histological necroinflammatory activity grade at *r*AlH diagnosis was grade 1 in 28 patients (19%), grade 2 in 100 patients (68%), grade 3 in 14 (10%), and grade 4 in 5 patients (3%); whereas the fibrosis stage at rAlH diagnosis was 0 in 20 patients (14%), stage 1 in 72 (49%), stage 2 in 40 (27%), 3 in 10 (7%) and stage 4 in 5 (3%).

Primary immunosuppression after LT included tacrolimus in 575 patients (78%), cyclosporine in 80 patients (11%), mycophenolate mofetil in 404 patients (55%), prednisone in 557 patients (76%), and azathioprine in 76 patients (10%) (Table 1).

The initial mean trough level of tacrolimus was 8.6 ng/ml (range, 1-40 ng/ml), and the initial mean trough level of cyclosporine was 327 ng/ml (range, 11-1,540 ng/ml). The mean initial dose of azathioprine was 100 mg daily (range, 50-250 mg/daily), mycophenolate mofetil was 1,523 mg daily (range, 360-1,200 mg/daily) and prednisone-prednisolone 21 mg/daily (range, 5-100 mg/daily).

Extended maintenance with prednisone or prednisolone was used in 362 patients (49%), for more than 1 year after LT, and in 170 patients (23%) more than 5 years after LT. Fifty-six patients had changes in their main immunosuppression after the first year that included switching from cyclosporine to tacrolimus (n = 13), tacrolimus to cyclosporine (n = 12), mycophenolate



Fig. 1. Cumulative probability of rAIH. The probability of *r*AIH was 20%, 31%, 37%, and 49% at 5-, 10-, 15-, and 20-years, respectively. Cumulative probabilities were calculated using the Kaplan-Meier method. *r*AIH, recurrent autoimmune hepatitis.

Table 2. Biochemical features associated with recurrent AIH after liver transplantation in univariable analyses.

Biochemical features	All patients	rAIH	No rAIH	HR	95% CI	p value
	(n = 736)	(n = 147)	<u>(n = 589)</u>			
Pre-LT						
ALT U/L	196±18	156±21	205±21	1.00	0.999-1.00	0.22
ALT times ULN	4.2±0.4	3.7±0.5	4.3±0.5	0.99	0.97-1.01	0.48
AST U/L	217±22	175±24	226±27	1.00	0.999-1.00	0.37
AST times ULN	5.5±0.5	5.1±0.8	5.6±0.7	0.997	0.98-1.01	0.70
Bilirubin µmol/L	189±13	216±44	183±12	1.00	1.00-1.001	0.56
Bilirubin times ULN	8.6±0.5	11.6±2.2	7.9±0.4	1.01	1.00-1.01	0.06
MELD Pre-LT	23±0.4	23±1	22±0.4	1.01	0.99-1.03	0.17
IgG g/L	23±0.5	27±2	22±0.5	1.03	1.01-1.04	0.006
IgG times ULN	1.4±0.03	1.7±0.1	1.4±0.03	1.47	1.10-1.95	0.008
3-month						
ALT U/L	53±4	64±12	50±4	1.001	1.00-1.002	0.19
ALT times ULN	1.2±0.08	1.4±0.2	1.1±0.1	1.04	0.97-1.12	0.27
AST U/L	37±2	46±7	35±2	1.002	1.00-1.01	0.05
AST times ULN	1.0±0.1	1.2±0.2	1.0±0.1	1.02	0.95-1.09	0.60
Bilirubin µmol/L	20±2	22±4	19±2	1.003	1.00-1.01	0.08
Bilirubin times ULN	0.9±0.1	1.0±0.2	0.9±0.1	1.01	0.95-1.08	0.74
6-month						
ALT U/L	55±4	63±10	52±5	1.001	1.00-1.002	0.07
ALT times ULN	1.2±0.1	1.4±0.2	1.1±0.1	1.06	1.01-1.11	0.03
AST U/L	39±3	44±6	38±3	1.003	1.00-1.01	0.04
AST times ULN	1.0±0.06	1.1±0.2	1.0±0.1	1.13	1.02-1.25	0.02
Bilirubin µmol/L	23±1	24±3	22±1	1.01	1.00-1.01	0.02
Bilirubin times ULN	1.1±0.1	1.2±0.1	1.1±0.1	1.19	1.05-1.34	0.006
12-month						
ALT U/L	46±4	63±8	41±4	1.001	1.00-1.003	0.01
ALT times ULN	1.0±0.07	1.4±0.2	0.9±0.1	1.08	1.03-1.14	0.004
AST U/L	37±2	48±6	34±2	1.003	1.001-1.01	0.003
AST times ULN	1.0±0.06	1.2±0.2	0.9±0.1	1.12	1.04-1.21	0.004
Bilirubin µmol/L	22±3	26±7	21±4	1.001	0.999-1.002	0.44
Bilirubin times ULN	1.1±0.2	1.3±0.4	1.0±0.2	1.01	0.98-1.05	0.50

AIH, autoimmune hepatitis; ALT, alanine aminotransferase; AST, aspartate aminotransferase; HR, hazard ratio; LT, liver transplantation; MELD, model for end-stage liver disease; rAIH, recurrent AIH; ULN, upper limit of normal. Hazard ratios were calculated using Cox proportional regression analyses.

mofetil to azathioprine (n = 19), and azathioprine to mycophenolate (n = 12).

Clinical features associated with autoimmune hepatitis recurrence

By univariate Cox proportional hazard regression analysis, younger age at diagnosis of AIH (HR 0.98, 95% CI 0.97-0.99, p <0.001) and at the time of LT (HR 0.98, 95% CI 0.97-0.99, p <0.001) were associated with a higher risk of *r*AIH. Patients younger than 34 years-old at diagnosis of AIH and younger

than 42 years-old at the time of LT had a higher risk for rAIH (HR 1.76, 95% CI 1.20-2.58, p = 0.004, and HR 1.91, 95% CI 1.34-2.72, p < 0.001; respectively; Table 1). The year of LT (HR 0.99, 95% CI 0.96-1.01, p = 0.28) and the LT center volume (HR 1.08, 95% CI 0.72-1.61, p = 0.72) were not significantly associated with rAIH.

The use of mycophenolate mofetil after LT (HR 1.59; 95% CI 1.11-2.27; p = 0.01), concomitant autoimmune disease (HR 1.56; 95% CI 1.07-2.26; p = 0.02), living-related LT (HR 0.35, 95% CI 0.18-0.68, p = 0.002), Roux-en-Y bile duct anastomosis (HR 1.93;

Table 3.	Features	associated	with AIH	recurrence by	y multivariable	analyses.
----------	----------	------------	----------	---------------	-----------------	-----------

	HR	95% CI	p value
Age at diagnosis ≤34 (years)	0.60	0.23-1.57	0.30
Age at LT ≤42 (years)	3.15	1.22-8.16	0.02
Concomitant autoimmune disease	1.50	0.76-2.94	0.24
Tacrolimus post-LT	1.91	0.27-13.39	0.52
Cyclosporine post-LT	1.52	0.22-10.38	0.67
Prednisone or prednisolone post-LT	0.77	0.35-1.67	0.50
Mycophenolate mofetil post-LT	3.06	1.39-6.73	0.005
Azathioprine post-LT	1.56	0.46-5.27	0.48
Rejection (acute)	1.60	0.88-2.93	0.13
Type of LT, living related	0.39	0.14-1.11	0.08
Bile duct anastomosis, Roux-en-Y	1.73	0.70-4.25	0.23
Donor/recipient sex mismatch	2.57	1.39-4.76	0.003
Bilirubin µmol/L (6-mo)	1.00	0.999-1.001	0.61
IgG g/L (pre-LT)	1.04	1.01-1.06	0.004
ALT U/L (12-mo)	1.00	0.998-1.003	0.74

Hazard ratios were calculated using Cox proportional regression analyses. AIH, autoimmune hepatitis; ALT, alanine aminotransferase; HR, hazard ratio; LT, liver transplantation.



Fig. 2. Cumulative probability of rAIH. (A) In patients younger and older than 42 years at transplantation. The 5-year probability of rAIH was 25% and 15%, respectively (*p* <0.001, log-rank test). The 10-year probability of survival was 40% and 26% in these same groups. (B) In patients with and without donor/recipient sex mismatch. The 5-year probability of rAIH was 26% and 17%, respectively (*p* = 0.04, log-rank test). The 10-year probability of survival was 40% and 26% in these same groups. (B) In patients with and without donor/recipient sex mismatch. The 5-year probability of rAIH was 26% and 17%, respectively (*p* = 0.04, log-rank test). The 10-year probability of survival was 40% and 28% in these

95% CI 1.17-3.17; p = 0.009), donor/recipient sex mismatch (HR 1.48; 95% CI 1.02-2.15; p = 0.04), and acute rejection episodes (HR 2.04; 95% CI 1.45-2.85; p < 0.001) were associated with rAIH (Table 1). There was no significant association with other clinical features such as recipient sex, ethnicity, age or sex of the donor, LT indication, changes in the main immunosuppression after the first year of LT, explant necroinflammatory activity, fibrosis, and infiltration with plasma cells and the risk of rAIH (Table 1).

In a sub-group analysis of sex mismatch, the frequency of male donor to female recipient mismatching was 78% (n = 230) and did not show a significant association with *r*AIH (HR 0.98, 95% CI 0.54-1.77, p = 0.95) when compared to female donor to male recipient mismatching.

Biochemical features associated with autoimmune hepatitis recurrence

By univariate Cox proportional hazard analysis, elevation of IgG pre-LT, bilirubin at 6 months, and both ALT and AST at 12 months after LT, were associated with a higher risk of rAIH (Table 2). Notably, a trend toward elevated ALT and AST levels from 3 months onwards, within the first year, was linked with increased risk of rAIH (Table 2), and more specifically, ALT above the ULN at 12 months after LT (HR 1.08, 95% CI 1.03-1.14, p = 0.004).

Multivariable analyses of features associated with autoimmune hepatitis recurrence

In the multivariable analysis, which included age at diagnosis \leq 34 years, age at LT \leq 42 years, concomitant autoimmune disease, post-LT use of tacrolimus, cyclosporine, mycophenolate mofetil and azathioprine, acute rejection, living-related LT, Roux-en-Y bile duct anastomosis, sex mismatch, bilirubin at 6 months, ALT at 12 months as well as IgG pre-LT, only age at LT \leq 42 years (HR, 3.15; 95% CI, 1.22-8.16; *p* = 0.02), use of mycophenolate mofetil (HR, 3.06; 95% CI, 1.39-6.73; *p* = 0.005), sex mismatch (HR, 2.57; 95% CI, 1.39-4.76; *p* = 0.003) and IgG pre-LT (HR, 1.04; 95% CI, 1.01-1.06; *p* = 0.004) were independently associated with *r*AIH (Table 3, Fig. 2A-D).

In addition, we performed a multivariate subanalysis including only patients who had a liver biopsy after LT (n = 529), that showed that use of mycophenolate mofetil, sex mismatch and IgG pre-LT were associated with a higher risk of *r*AIH (Table S1).

Patient and graft survival associated with recurrent disease

Overall median survival after LT was 23 (IQR: 9-27) years. The overall 5-, 10-, 15- and 20-year probability of survival was 86%, 72%, 65%, and 58%, respectively (Fig. 3A). The graft 5-, 10-, 15-, and 20-year probability of graft survival was 88%, 81%, 75%, and 68%, respectively (Fig. 3B).

In a Cox proportional hazard regression analysis implementing recurrence as a time-dependent covariate, *r*AIH (HR 9.61, 95% Cl 5.33-17.30, *p* <0.001) was associated with graft failure. Sex (HR 1.06, 95% Cl 1.00-1.12, *p* = 0.045), use of mycophenolate mofetil (HR 1.10, 95% Cl 1.04-1.17, *p* = 0.001), acute rejection (HR 1.11, 95% Cl 1.05-1.17, *p* <0.001) and bilirubin at 12-months post-LT (HR

1.004, 95% CI 1.002-1.01, p < 0.001) were also associated with graft failure in univariate analysis. However, only bilirubin at 12-months post-LT (HR 1.004, 95% CI 1.002-1.01, p < 0.001) and rAIH (time-dependent HR 10.79, 95% CI 5.37-21.66, p < 0.001, Table 4) were independently associated with graft failure in the multivariate analysis.

In the univariate Cox regression analysis, *r*AIH, as a timedependent covariate (HR 1.95, 95% CI 1.24-3.06, *p* = 0.004), age at LT ≤42 years (HR 0.96, 95% CI 0.92-0.996, *p* = 0.03), bilirubin at 12-months post-LT (HR 1.003, 95% CI 1.001-1.01, *p* <0.001) and ALT at 12-months post-LT (HR 1.002, 95% CI 1.00-1.003, *p* = 0.002) were all associated with overall survival after LT. However, in the multivariable analysis, bilirubin at 12-months post-LT (HR 1.003, 95% CI 1.002-1.01, *p* <0.001), ALT at 12-months post-LT (HR 1.002, 95% CI 1.002-1.003, *p* = 0.004) and *r*AIH (HR 2.53, 95% CI 1.48-4.33, *p* = 0.001) were independently associated with overall survival after LT (Table 4).

In the multivariable Cox regression analysis, including only those patients who had liver biopsies after LT (n = 529), results were similar (Table S2).

Graft survival was significantly diminished to 12.2 years (95% CI 10.7-13.6) in patients with *r*AIH compared to 24.0 years (95% CI 21.4-26.6) in patients without *r*AIH (p < 0.001, Fig. 4A).

The 5-, 10-, 15- and 20-year probability of graft survival was 78%, 65%, 53% and 53% in patients with rAIH and 96%, 93%, 93%, and 87% in patients without rAIH (Log-rank, p < 0.001, Fig. 4A). The majority of patients with recurrence of AIH lost their graft as a result of cirrhosis related to rAIH (72%), and the remainder were attributed to either rejection (14%) or hepatic artery thrombosis-ischemic cholangiopathy (11%) and other etiologies (3%). In contrast, patients without recurrence of AIH lost their allograft as a result of rejection (23%), hepatic artery thrombosis-ischemic cholangiopathy (43%), or other causes (34%).

The overall survival was 15.1 years (95% CI 12.7-17.5) for patients with *r*AIH compared to 20.0 years (95% CI 17.8-22.2) for those without recurrent disease (p < 0.001; Fig. 4B). The 5-, 10-, 15- and 20-year probability of overall survival was 81%, 73%, 55%, and 44% in patients with *r*AIH and 93%, 81%, 75%, and 61% in patients without *r*AIH (Log-rank, p < 0.001, Fig. 4B).

The increased risk of death after LT in AIH was mainly attributable to infections, such as fungal sepsis occurring soon after LT. In addition, we found that the risk of septic complications after LT was higher in patients receiving mycophenolate mofetil (HR, 1.92, 95% CI 1.28-2.90, p = 0.002), whereas the use of azathioprine was associated with a lower risk (HR, 0.48, 95% CI 0.26-0.87, p = 0.02).

Discussion

In the largest comprehensive cohort of liver transplant patients with AIH to date, we demonstrate that recurrent disease affects more than one-third of patients at 10 years after LT. Furthermore, recurrence significantly increases the risk of graft loss and mortality, providing an impetus to evaluate potential interventions to prevent *r*AIH.⁹ The type of immunosuppression after LT, specifically the use of mycophenolate mofetil, in

same groups. (C) In patients receiving mycophenolate mofetil after liver transplantation. The 5-year probability of rAIH was 24% and 16%, respectively (p = 0.01, log-rank test). The 10-year probability of survival was 36% and 27% in these same groups. (D) In patients with IgG >20 and ≤ 20 g/L before liver transplantation. The 5-year probability of rAIH was 24% and 14%, respectively (p = 0.05, log-rank test). The 10-year probability of survival was 37% and 22% in these same groups. (D) In patients with IgG >20 and ≤ 20 g/L before liver transplantation. The 5-year probability of rAIH was 24% and 14%, respectively (p = 0.05, log-rank test). The 10-year probability of survival was 37% and 22% in these same groups. Cumulative incidences were calculated using the Kaplan-Meier method, and they were compared using the Log-rank (Mantel-Cox) test. rAIH, recurrent auto-immune hepatitis.



Fig. 3. Survival of patients with rAlH after liver transplantation. (A) Overall survival. The 5-, 10-, 15- and 20-year probability of survival was 86%, 72%, 65%, and 58%, respectively. (B) Graft Survival. The 5-, 10-, 15-, and 20-year graft survival probability was 88%, 81%, 75%, and 68%, respectively. Cumulative probabilities were calculated using the Kaplan-Meier method. LT, liver transplantation; *r*AlH, recurrent autoimmune hepatitis.

addition to younger age at LT, donor/recipient sex mismatch and high IgG level before LT were also found to be associated with the higher risk of rAIH. Prior single-center studies reported that rAIH had no significant impact on long-term survival or the need for re-transplantation.^{6,9} However, the major limitations of these studies included the small size and their lack of long-term follow-up following LT, thereby limiting the probability of detecting differences in clinical outcomes. In addition, these case series were likely too small to demonstrate an increased risk of graft loss or mortality in recipients with rAIH compared to recipients with other liver diseases. Nevertheless, progression to cirrhosis, graft failure, and retransplantation have previously been reported with rAIH^{20,21} and these outcomes, estimated at a range of 13-23% in smaller series, have encouraged attempts to make an earlier diagnosis and find more effective treatment.

We also found that the probability of *r*AIH approaches 50% at 20 years. Indeed, we may have underestimated the incidence of recurrence in this study because not all patients underwent protocol liver biopsies necessary for confirmation of the diagnosis. In addition, younger age at LT was associated with a higher risk of *r*AIH, which is also a risk factor for graft loss and mortality. This finding is in agreement with other studies suggesting that age of onset and LT may be associated with a more aggressive AIH phenotype.^{22,23}

This study provides further insight regarding the use of immunosuppression after LT with respect to both the incidence and accelerated onset of *r*AIH. We found that patients receiving mycophenolate had an increased risk of *r*AIH. Interestingly, there were no significant differences in the initial dose of mycophenolate mofetil (1,519±60 vs. 1,533±48 mg/daily, p = 0.97), nor the dose at 1 year after LT (1,335±76 vs. 1,314±41 mg/daily, p = 0.82) among

Table 4. Features associated with graft and patient survival after liver transplantation.

		Univariate			Multivariab	le
Features	HR	95% CI	p value	HR	95% CI	p value
Graft survival analysis						
Sex, male	1.06	1.00-1.12	0.045	1.56	0.80-3.05	0.20
Age at diagnosis ≤34 (years)	1.02	0.97-1.08	0.43			
Age at LT ≤42 (years)	1.05	0.997-1.11	0.06			
Mycophenolate mofetil post-LT	1.10	1.04-1.17	0.001	1.69	0.90-3.19	0.10
Rejection (acute)	1.11	1.05-1.17	< 0.001	1.42	0.77-2.61	0.27
Recurrence of AIH**	9.61	5.33-17.30	<0.001	10.79	5.37-21.66	<0.001
ALT U/L (12-mo)	1.002	1.00-1.004	0.10			
Bilirubin µmol/L (12-mo)	1.004	1.002-1.01	< 0.001	1.004	1.002-1.01	< 0.001
Overall survival analysis						
Sex, male	1.02	0.97-1.06	0.47			
Age at diagnosis ≤34 (years)	0.96	0.92-1.01	0.09			
Age at LT ≤42 (years)	0.96	0.92-0.996	0.03	0.62	0.37-1.02	0.06
Mycophenolate mofetil post-LT	1.03	0.99-1.08	0.13			
Rejection (acute)	1.03	0.995-1.07	0.09			
Recurrence of AIH**	1.95	1.24-3.06	0.004	2.53	1.48-4.33	0.001
ALT U/L (12-mo)	1.002	1.00-1.003	0.002	1.002	1.00-1.003	0.004
Bilirubin µmol/L (12-mo)	1.003	1.001-1.01	<0.001	1.003	1.002-1.01	<0.001

Hazard ratios were calculated using Cox regression analyses with time-dependent covariate. AIH, autoimmune hepatitis; ALT, alanine aminotransferase; HR, hazard ratio; LT, liver transplantation.

**These hazard ratios were obtained by considering recurrent AIH as a time-dependent covariate in univariable and multivariable analyses.

patients with and without rAIH. This finding suggests that we should consider the use of other agents such as azathioprine, mercaptopurine or long-term low dose of steroids in addition to calcineurin inhibitors. Indeed, prior studies have reported that patients receiving long-term prednisone have a lower risk of recurrence.⁹ A previous study from the ELTR revealed inferior outcomes in patients transplanted for AIH compared to those transplanted for PBC or PSC.¹⁰ The increased risk of death after LT in AIH was mainly attributable to infections, such as fungal sepsis occurring soon after LT. In fact, we found that the risk of septic complications after LT was higher in patients receiving mycophenolate mofetil, a potent immunosuppressive agent (HR 1.92, p =0.002), whereas azathioprine was associated with a lower risk (HR 0.48, p = 0.02). However, there are alternative hypotheses to explain differences in the risk of rAIH related to the use of separate immunosuppressive regimens. Some have argued for an "era effect" as the use of azathioprine and prednisone was more prevalent in the 1980s and 1990s, when other factors such as cold ischemia times, shorter waiting periods and fewer sick patients might have impacted the development of rAIH. The role of mycophenolate mofetil on adaptive immune cells is also worth investigating in the future.

Another finding from our study is that donor and recipient sex mismatch was associated with a higher risk of *r*AIH. While prior studies have reported that recipients of sex-mismatched allografts had an 11% higher risk of graft loss,²⁴ none have linked this association with *r*AIH, possibly due to insufficient sample sizes in most reports.^{6,21,25} Estrogen has a significant impact on immune cells and creating a tolerogenic environment; thus, this should be explored in future research.²⁶

Similar to others,⁶ we found that high IgG levels before LT were associated with a higher risk of rAIH. In fact, patients with IgG levels in excess of 20 g/L had an increased risk of rAIH (HR 1.76, 95% CI 1.11-2.79, p = 0.02) and this observation should prompt increased vigilance for rAIH because the disease may reoccur at any time after transplantation.

Another novel observation was that early elevation of ALT and the subsequent elevations in AST and bilirubin as well, within the first 12 months following LT (Tables 2 & 3), were helpful in risk stratification of *r*AIH. Although we lack biomarkers to distinguish whether the presence of high ALT at 6 and 12 months constitutes a risk factor for subsequent development of *r*AIH or indicates an early manifestation of *r*AIH, similar observations of early biochemical changes predicting subsequent recurrent disease in patients with PBC and PSC have also recently been reported.^{12,27} Indeed, the use of more potent immunosuppression within the first year may mask the characteristic histological presentation of AIH, whereas non-specific inflammatory changes are a more common finding early on in the disease process.

Notably, 10% of patients had advanced fibrosis (F3-4) at the time of *r*AIH diagnosis. Even though there was no difference in the frequency of advanced fibrosis (F3-F4) among patients with protocol or clinically indicated biopsies (4% vs. 12%, p = 0.30), this could be related to the small number of patients (low power) and we recommend the implementation of protocol liver biopsies in patients transplanted for AIH, especially if they have factors associated with a higher risk of *r*AIH (younger age at LT, sex mismatch and high IgG pre-LT).

The pathogenic mechanisms for AIH before LT are probably similar to those that promote *r*AIH. However, the introduction of a donor allograft with different antigen-presenting and immunereactive cells will likely modulate the immune response. Autoimmunity infers that there is a loss of tolerance for self-antigens and that an immune response is mounted against host antigens presented by host-derived antigen-presenting cells.^{28,29} After LT, autoantigens may not only be presented by self-derived class I and II HLAs and recognized by self-derived CD4+ and CD8+ T cells, but they may also be presented by donor-derived HLAs and trigger an alloimmune response indistinguishable from autoimmunity. Accordingly, recipient memory T cells restricted to selfderived HLAs may also react with donor-derived HLAs, and trigger *r*AIH as a consequence.



Fig. 4. Survival in patients with and without rAIH after liver transplantation. (A) Graft survival. (B) Overall survival. Patients who had no rAIH during their follow-up are in the blue line. Patients who developed rAIH are only represented in the blue line until they developed rAIH. These patients are censored and switched to a new survival curve (red line) once they have rAIH. The time is then reset as time 0 for their further follow-up. Levels of significance: *p* <0.001 [the semi-Markov models ("clock reset" model)]. LT, liver transplantation; rAIH, recurrent autoimmune hepatitis.

Most cases of rAIH in this cohort (93%) occurred in patients receiving calcineurin inhibitors, such as tacrolimus or cyclosporine. These drugs block T-cell activation through T-cell receptors, where interleukin-2 (IL-2) is required for the survival and proliferation of regulatory T cells (Tregs) which express the IL-2 receptor, CD25.³⁰ As these medications decrease the production of IL-2, Treg function may be altered. In animal models, autoimmune disease linked with the use of these immunosuppressive regimens has been related to the impairment of T cell-regulated suppressor function and subsequent development and activation of autoreactive T cells.^{31,32} Thus, manipulation of Tregs following LT with IL-2-directed therapy may be relevant to prevent rAIH in the allograft.³³

Our results suggest that patients at higher risk of *r*AIH should be considered for therapeutic strategies within the first year of LT to prevent the occurrence of *r*AIH. Treatment of *r*AIH is empiric and depends on the presentation, which can be variable. When patients present with asymptomatic disease and minimal changes in liver tests or histological features, minor increases in immunosuppression may be sufficient to suppress recurrent disease.^{34,35}

Of note and similar to previous studies in patients with AIH, the level of bilirubin at diagnosis of *r*AIH was associated with a higher risk of graft loss (HR 1.004, 95% CI 1.002-1.006, *p* <0.001) and death (HR 1.003, 95% CI 1.001-1.005, *p* = 0.04).^{36,37} In fact, patients with bilirubin >50 µmol/L at diagnosis of *r*AIH had a higher risk for graft loss (HR 4.21, 95% CI 2.13-8.34, *p* <0.001).

There was no difference in bilirubin levels at the time of rAIH between patients with advanced fibrosis (F3-4) compared to patients without advanced fibrosis (4.1±1.9 vs. 2.5±0.5 ULN, p = 0.31) suggesting that this finding was not related to missed cases of rAIH with advanced fibrosis.

A wide range in frequency of *r*AIH was observed, varying from 0.0% to 69% among centers, with a yearly recurrence rate ranging from 0% to 10%. This could be related to (i) different follow-up times between centers (mean total follow-up range 5-136 months, p < 0.001), and (ii) difference in protocol liver biopsies that were not performed in 10 of the 33 LT centers.

We acknowledge there are limitations in this study. Due to the retrospective nature of the study, we were not able to accurately evaluate immunosuppression adherence and time era effect. In addition, liver biopsies were not re-evaluated to identify other important aspects, such as the presence of signs of acute or chronic rejection. As well, different follow-up strategies, and changes of immunosuppression over time that were not recorded might have influenced our results. Moreover, while the diagnosis of rAIH in our cohort was established according to liver biopsies,^{38,39} biopsies were protocol-driven in some centers and clinically driven in others. This may have led to differences in time to diagnosis of rAIH between the different centers. Indeed, in the Cox regression analysis, clinically driven liver biopsies were associated with a higher risk of rAIH (Table 1), which may have been related to the higher frequency of abnormal serum liver function tests in patients with rAIH. Moreover, the mean time for rAIH was different between centers that perform protocol- and clinically driven liver biopsies (16.3 years, 95% CI 14.9-17.8 vs. 14.5 years, 95% CI 12.6-16.3 years, p <0.001). In addition, lack of biopsies after LT (especially protocol biopsies) might lead to overestimation of the prognosis after rAIH since mild cases may have been overseen or misclassified as not having rAIH. In order to minimize this, we performed a subanalysis in patients who had liver biopsies after LT (n = 529), and we found similar results; however, we recognized that the "true" rAIH prevalence and effect of the recurrence of the disease on prognosis would be better reflected in patients with protocol liver biopsies. Yet, we decided not to perform a subanalysis on patients with protocol biopsies, due to the small number of patients (n = 204), and the low number of events (rAIH: 27, graft failure: 4, and deaths: 25), which would not be representative of the whole cohort.

Although pre-screening of variables based on univariable p values is a commonly accepted approach and has been adopted in our large multicenter study, this strategy might lead to overestimation and risks for optimism bias.⁴⁰ Variables with p value equal or less than 0.05 in the univariate analysis were included in the Cox regression multivariate analysis to avoid model over-fitting. In addition, we acknowledge that handling missing data using imputation by the mean or using a specific category for missing data is known to be suboptimal; however, as we had a relatively low frequency of missing data, we consider that the use of these strategies did not have a significant effect on our results.

Importantly, we found a high variation between centers regarding the frequency of *r*AlH (range, 0% to 69%) and yearly recurrence rate (range, 0% to 10.1%). This broad variation seems to reflect differences in vigilance after LT, indication for liver biopsies, and different treatment strategies, emphasizing the need of standardized guidelines for follow-up and management strategies in patients transplanted for AIH.

In conclusion, we found that a younger age at the time of LT, use of mycophenolate mofetil, donor/recipient sex mismatch and IgG pre-LT were independently associated with an increased risk of rAIH in this large cohort of LT recipients with AIH. Recurrent AIH was associated with worse graft and overall survival after LT, suggesting that exploration of therapeutic interventions to prevent and treat rAIH are warranted.

Abbreviations

AIH, autoimmune hepatitis, ALT, alanine aminotransferase; AST, aspartate aminotransferase; ELTR, European Liver Transplant Registry; HR, hazard ratio; INR, international normalized ratio; LT, liver transplantation; PBC, primary biliary cholangitis; PSC, primary sclerosing cholangitis; rAIH, recurrent AIH; Treg, regulatory T cell; ULN, upper limit of normal; UNOS, United Network for Organ Sharing.

Financial support

Aldo J. Montano-Loza receives funding from the University of Alberta Hospital Foundation (UHF) and the Canadian Liver Foundation (CLF). Maryam Ebadi receives funding from the Canadian Institutes of Health Research (CIHR) - Institute of Nutrition, Metabolism, and Diabetes (INMD) Fellowship–Hepatology, in partnership with the Canadian Association for the Study of the Liver (CASL) and the Canadian Liver Foundation (CLF). Ye Htun Oo receives funding from Sir Jules Thorn Biomedical Research Award, Transbioline Innovative Medicine Initiatives, Medical Research Foundation, MRC, Queen Elizabeth Hospital Birmingham Charity and EASL. Vincenzo Ronca receives funding from EASL Juan Rodes PhD fellowship programme.

Conflicts of interest

These authors disclose the following: A.J. Montano-Loza has served on advisory boards for Intercept Pharmaceuticals. B.E. Hansen reports grants from Intercept Pharmaceuticals and Zambon Nederland B.V. and consulting work for Intercept Pharmaceuticals and Novartis. A.E. Kremer reports consulting work for CymaBay, GSK, Intercept Pharmaceuticals, and Mirum and grants from Intercept Pharmaceuticals. A. Parés consults for Intercept and Novartis. A. Floreani reports consulting activities for Intercept Pharmaceuticals. A. Mason consults for, is on the speakers' bureau of, and received grants from Intercept. He received grants from Merk. The remaining authors disclose no conflicts.

Please refer to the accompanying ICMJE disclosure forms for further details.

Authors' contributions

Aldo J. Montano-Loza, study design, analyses of the data, creation of the first draft of the manuscript and final version; Vincenzo Ronca, analyses of the data, creation of the first draft of the manuscript and final version; Maryam Ebadi, analyses of the data, creation of the first draft of the manuscript and final version and submitting manuscript for review; Bettina E. Hansen, analyses of the data, creation of the first draft of the manuscript and final version; Gideon Hirschfield, analyses of the data, creation of the first draft of the manuscript and final version; Saleh Elwir, analysis of the data and critical revision of the manuscript for important intellectual content; Mohamad Alsaed, analysis of the data and critical revision of the manuscript for important intellectual content; Piotr Milkiewicz, analysis of the data and critical revision of the manuscript for important intellectual content; Maciej Janik, analysis of the data and critical revision of the manuscript for important intellectual content: Hanns-Ulrich Marschall, analysis of the data and critical revision of the manuscript for important intellectual content; Maria Antonella Burza, analysis of the data and critical revision of the manuscript for important intellectual content; Cumali Efe, analysis of the data and critical revision of the manuscript for important intellectual content; Ali Rıza Calışkan, analysis of the data and critical revision of the manuscript for important intellectual content; Murat Harputoğlu, analysis of the data and critical revision of the manuscript for important intellectual content; Gökhan Kabaçam, analysis of the data and critical revision of the manuscript for important intellectual content; Débora Terrabuio, analysis of the data and critical revision of the manuscript for important intellectual content; Fernanda de Quadros Onofrio, analysis of the data and critical revision of the manuscript for important intellectual content; Alan Bonder, analysis of the data and critical revision of the manuscript for important intellectual content; Albert Parés, analysis of the data and critical revision of the manuscript for important intellectual content; Laura Llovet, analysis of the data and critical revision of the manuscript for important intellectual content; Murat Akyıldız, analysis of the data and critical revision of the manuscript for important intellectual content; Cigdem Arikan, analysis of the data and critical revision of the manuscript for important intellectual content; Michael P. Manns, analysis of the data and critical revision of the manuscript for important intellectual content; Richard Taubert, analysis of the data and critical revision of the manuscript for important intellectual content; Anna-Lena Weber, analysis of the data and critical revision of the manuscript for important intellectual content; Thomas D Schiano, analysis of the data and critical revision of the manuscript for important intellectual content; Brandy Haydel, analysis of the data and critical revision of the manuscript for important intellectual content; Piotr Czubkowski, analysis of the data and critical revision of the manuscript for important intellectual content; Piotr Socha, analysis of the data and critical revision of the manuscript for important intellectual content; Natalia Ołdak, analysis of the data and critical revision of the manuscript for important intellectual content; Nobuhisa Akamatsu, analysis of the data and critical revision of the manuscript for important intellectual content; Atsushi Tanaka, analysis of the data and critical revision of the manuscript for important intellectual content; Cynthia Levy, analysis of the data and critical revision of the manuscript for important intellectual content; Eric F. Martin, analysis of the data and critical revision of the manuscript for important intellectual content; Aparna Goel, analysis of the data and critical revision of the manuscript for important intellectual content; Mai Sedki, analysis of the data and critical revision of the manuscript for important intellectual content; Irena Jankowska, analysis of the data and critical revision of the manuscript for important intellectual content; Toru Ikegami, analysis of the data and critical revision of the manuscript for important intellectual content; Maria Rodriguez, analysis of the data and critical revision of the manuscript for important intellectual content; Martina Sterneck, analysis of the data and critical revision of the manuscript for important intellectual content; Christina Weiler-Normann, analysis of the data and critical revision of the manuscript for important intellectual content; Christoph Schramm, analysis of the data and critical revision of

the manuscript for important intellectual content; Maria Francesca Donato, analysis of the data and critical revision of the manuscript for important intellectual content; Ansgar Lohse, analysis of the data and critical revision of the manuscript for important intellectual content; Raul J. Andrade, analysis of the data and critical revision of the manuscript for important intellectual content; Vilas R. Patwardhan, analysis of the data and critical revision of the manuscript for important intellectual content; Bart van Hoek, analysis of the data and critical revision of the manuscript for important intellectual content; Maaike Biewenga, analysis of the data and critical revision of the manuscript for important intellectual content; Andreas E. Kremer, analysis of the data and critical revision of the manuscript for important intellectual content; Yoshihide Ueda, analysis of the data and critical revision of the manuscript for important intellectual content; Mark Deneau, analysis of the data and critical revision of the manuscript for important intellectual content; Mark Pedersen, analysis of the data and critical revision of the manuscript for important intellectual content; Marlyn J. Mayo, analysis of the data and critical revision of the manuscript for important intellectual content; Annarosa Floreani, analysis of the data and critical revision of the manuscript for important intellectual content; Patrizia Burra, analysis of the data and critical revision of the manuscript for important intellectual content; Maria Francesca Secchi, analysis of the data and critical revision of the manuscript for important intellectual content; Benedetta Terziroli Beretta-Piccoli, analysis of the data and critical revision of the manuscript for important intellectual content; Marco Sciveres, analysis of the data and critical revision of the manuscript for important intellectual content; Giuseppe Maggiore, analysis of the data and critical revision of the manuscript for important intellectual content; Syed-Mohammed Jafri, analysis of the data and critical revision of the manuscript for important intellectual content; Dominique Debray, analysis of the data and critical revision of the manuscript for important intellectual content; Muriel Girand, analysis of the data and critical revision of the manuscript for important intellectual content; Florence Lacaille, analysis of the data and critical revision of the manuscript for important intellectual content; Ellina Lytvyak, collection of data across all center, analysis of the data and critical revision of the manuscript for important intellectual content; Andrew L. Mason, analysis of the data and critical revision of the manuscript for important intellectual content; Michael Heneghan, study design, analyses of the data, creation of the first draft of the manuscript and final version; Ye Htun Oo study design, analyses of the data, creation of the first draft of the manuscript and final version.

Data availability statement

The datasets generated and analyzed during the current study are not publicly available but are available from the corresponding author on reasonable request.

Acknowledgement

Miguel Jiménez-Perez, MD kindly provided information of patients transplanted at the Regional Hospital of Málaga, Spain.

Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.jhep.2022.01.022.

References

Author names in bold designate shared co-first authorship

- [1] Mieli-Vergani G, Vergani D, Czaja AJ, Manns MP, Krawitt EL, Vierling JM, et al. Autoimmune Hepatitis. Nat Rev Dis Primers 2018;4:18017.
- [2] Lohse AW, Chazouilleres O, Dalekos G, Drenth JPH, Heneghan M, Hofer H, et al. EASL clinical practice guidelines: autoimmune hepatitis. J Hepatol 2015;63:971–1004.
- [3] Mendes F, Couto CA, Levy C. Recurrent and de novo autoimmune liver diseases. Clin Liver Dis 2011;15:859–878.
- [4] Stirnimann G, Ebadi M, Czaja AJ, Montano-Loza AJ. Recurrent and de novo autoimmune hepatitis. Liver Transpl 2019;25:152–166.
- [5] Montano-Loza AJ, Bhanji RA, Wasilenko S, Mason AL. Systematic review: recurrent autoimmune liver diseases after liver transplantation. Aliment Pharmacol Ther 2017;45:485–500.
- [6] Montano-Loza AJ, Mason AL, Ma M, Bastiampillai RJ, Bain VG, Tandon P. Risk factors for recurrence of autoimmune hepatitis after liver transplantation. Liver Transpl 2009;15:1254–1261.
- [7] Montano-Loza AJ, Wasilenko S, Bintner J, Mason AL. Cyclosporine A protects against primary biliary cirrhosis recurrence after liver transplantation. Am J Transplant Off J Am Soc Transplant Am Soc Transpl Surgeons 2010;10:852–858.
- [8] Duclos-Vallee JC, Sebagh M. Recurrence of autoimmune disease, primary sclerosing cholangitis, primary biliary cirrhosis, and autoimmune hepatitis after liver transplantation. Liver Transpl 2009;15(Suppl 2):S25–S34.
- [9] Krishnamoorthy TL, Miezynska-Kurtycz J, Hodson J, Gunson BK, Neuberger J, Milkiewicz P, et al. Longterm corticosteroid use after liver transplantation for autoimmune hepatitis is safe and associated with a lower incidence of recurrent disease. Liver Transpl 2016;22:34–41.
- [10] Heinemann M, Adam R, Berenguer M, Mirza D, Malek-Hosseini SA, O'Grady JG, et al. Longterm survival after liver transplantation for autoimmune hepatitis: results from the European liver transplant Registry. Liver Transpl 2020;26:866–877.
- [11] Mack CL, Adams D, Assis DN, Kerkar N, Manns MP, Mayo MJ, et al. Diagnosis and management of autoimmune hepatitis in adults and children: 2019 practice guidance and guidelines from the American Association for the study of liver diseases. Hepatology 2020;72:671–722.
- [12] Mason AL, Montano-Loza AJ. Systematic investigation of elevated cholestatic enzymes during the third posttransplant month. Liver Transpl 2013;19:S23–S30.
- [13] Hennes EM, Zeniya M, Czaja AJ, Pares A, Dalekos GN, Krawitt EL, et al. Simplified criteria for the diagnosis of autoimmune hepatitis. Hepatology 2008;48:169–176.
- [14] Manns MP, Czaja AJ, Gorham JD, Krawitt EL, Mieli-Vergani G, Vergani D, et al. Diagnosis and management of autoimmune hepatitis. Hepatology 2010;51:2193–2213.
- [15] Kamath PS, Kim WR. The model for end-stage liver disease (MELD). Hepatology 2007;45:797–805.
- [16] Batts KP, Ludwig J. Chronic hepatitis. An update on terminology and reporting. Am J Surg Pathol 1995;19:1409–1417.
- [17] Holford T. Life tables with concomitant information. Biometrics 1976;32: 587–597.
- [18] Peto R, Pike MC. Conservatism of the approximation (0-E)/E in the log rank test for survival data on tumor incidence data. Biometrics 1973;29:579–584.
- [19] Andersen PK, Esbjerg S, Sorensen TI. Multi-state models for bleeding episodes and mortality in liver cirrhosis. Stat Med 2000;19:587–599.
- [20] Birnbaum AH, Benkov KJ, Pittman NS, McFarlaneFerreira Y, Rosh JR, LeLeiko NS. Recurrence of autoimmune hepatitis in children after liver transplantation. J Pediatr Gastroenterol Nutr 1997;25:20–25.

- [21] Reich DJ, Fiel I, Guarrera JV, Emre S, Guy SR, Schwartz ME, et al. Liver transplantation for autoimmune hepatitis. Hepatology 2000;32:693–700.
- [22] Czaja AJ, Carpenter HA. Distinctive clinical phenotype and treatment outcome of type 1 autoimmune hepatitis in the elderly. Hepatology 2006;43:532–538.
- [23] Czaja AJ, Strettell MD, Thomson LJ, Santrach PJ, Moore SB, Donaldson PT, et al. Associations between alleles of the major histocompatibility complex and type 1 autoimmune hepatitis. Hepatology 1997;25:317–323.
- [24] Lai JC, Feng S, Roberts JP, Terrault NA. Gender differences in liver donor quality are predictive of graft loss. Am J Transpl 2011;11:296–302.
- [25] Gonzalez-Koch A, Czaja AJ, Carpenter HA, Roberts SK, Charlton MR, Porayko MK, et al. Recurrent autoimmune hepatitis after orthotopic liver transplantation. Liver Transpl 2001;7:302–310.
- [26] Bozward AG, Wootton GE, Podstawka O, Oo YH. Autoimmune hepatitis: tolerogenic immunological state during pregnancy and immune escape in post-partum. Front Immunol 2020;11:591380.
- [27] Montano-Loza AJ, Hansen BE, Corpechot C, Roccarina D, Thorburn D, Trivedi P, et al. Factors associated with recurrence of primary biliary cholangitis after liver transplantation and effects on graft and patient survival. Gastroenterology 2019;156:96–107 e101.
- [28] Guido M, Burra P. De novo autoimmune hepatitis after liver transplantation. Semin Liver Dis 2011;31:71–81.
- [29] Oo YH, Sakaguchi S. Regulatory T-cell directed therapies in liver diseases. J Hepatol 2013;59:1127–1134.
- [30] Chen YY, Jeffery HC, Hunter S, Bhogal R, Birtwistle J, Braitch MK, et al. Human intrahepatic regulatory T cells are functional, require IL-2 from effector cells for survival, and are susceptible to fas ligand-mediated apoptosis. Hepatology 2016;64:138–150.
- [31] Bucy RP, Xu XY, Li J, Huang G. Cyclosporin A-induced autoimmune disease in mice. J Immunol 1993;151:1039–1050.
- [32] Kang HG, Zhang D, Degauque N, Mariat C, Alexopoulos S, Zheng XX. Effects of cyclosporine on transplant tolerance: the role of IL-2. Am J Transpl 2007;7:1907–1916.
- [33] Jeffery HC, Jeffery LE, Lutz P, Corrigan M, Webb GJ, Hirschfield GM, et al. Low-dose interleukin-2 promotes STAT-5 phosphorylation, T-reg survival and CTLA-4-dependent function in autoimmune liver diseases. Clin Exp Immunol 2017;188:394–411.
- [34] Prados E, Cuervas-Mons V, de la Mata M, Fraga E, Rimola A, Prieto M, et al. Outcome of autoimmune hepatitis after liver transplantation. Transplantation 1998;66:1645–1650.
- [35] Khalaf H, Mourad W, El-Sheikh Y, Abdo A, Helmy A, Medhat Y, et al. Liver transplantation for autoimmune hepatitis: a single-center experience. Transplant Proc 2007;39:1166–1170.
- [36] Montano-Loza AJ, Carpenter HA, Czaja AJ. Features associated with treatment failure in type 1 autoimmune hepatitis and predictive value of the model of end-stage liver disease. Hepatology 2007;46:1138–1145.
- [37] Yeoman AD, Westbrook RH, Zen Y, Maninchedda P, Portmann BC, Devlin J, et al. Early predictors of corticosteroid treatment failure in icteric presentations of autoimmune hepatitis. Hepatology 2011;53:926–934.
- [38] Ayata G, Gordon FD, Lewis WD, Pomfret E, Pomposelli JJ, Jenkins RL, et al. Liver transplantation for autoimmune hepatitis: a long-term pathologic study. Hepatology 2000;32:185–192.
- [39] Banff Working G, Demetris AJ, Adeyi O, Bellamy CO, Clouston A, Charlotte F, et al. Liver biopsy interpretation for causes of late liver allograft dysfunction. Hepatology 2006;44:489–501.
- [40] Riley RD, Ensor J, Snell KIE, Harrell Jr FE, Martin GP, Reitsma JB, et al. Calculating the sample size required for developing a clinical prediction model. BMJ 2020;368:m441.